Dear Reviewers:

We are very grateful to Reviewer for reviewing the paper so carefully. We have carefully considered the suggestion of Reviewer and make some changes. Responds to the reviewers' comments: “A 60-year-old man” is the most accurate expression, and we have further evaluated and modified the accuracy of the format and language of the article. Thank you for your guidance and contribution to the article.